



University of Groningen

Discovering Genetic Interactions in Large-Scale Association Studies by Stage-wise Likelihood Ratio Tests

Franberg, Mattias; Gertow, Karl; Hamsten, Anders; Lagergren, Jens; Sennblad, Bengt; PROCARDIS Consortium

Published in: PLoS genetics

DOI: 10.1371/journal.pgen.1005502

IMPORTANT NOTE: You are advised to consult the publisher's version (publisher's PDF) if you wish to cite from it. Please check the document version below.

Document Version Publisher's PDF, also known as Version of record

Publication date: 2015

Link to publication in University of Groningen/UMCG research database

Citation for published version (APA): Franberg, M., Gertow, K., Hamsten, A., Lagergren, J., Sennblad, B., & PROCARDIS Consortium (2015). Discovering Genetic Interactions in Large-Scale Association Studies by Stage-wise Likelihood Ratio Tests. *PLoS genetics*, *11*(9), [1005502]. https://doi.org/10.1371/journal.pgen.1005502

Copyright

Other than for strictly personal use, it is not permitted to download or to forward/distribute the text or part of it without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license (like Creative Commons).

The publication may also be distributed here under the terms of Article 25fa of the Dutch Copyright Act, indicated by the "Taverne" license. More information can be found on the University of Groningen website: https://www.rug.nl/library/open-access/self-archiving-pure/taverneamendment.

Take-down policy

If you believe that this document breaches copyright please contact us providing details, and we will remove access to the work immediately and investigate your claim.

Downloaded from the University of Groningen/UMCG research database (Pure): http://www.rug.nl/research/portal. For technical reasons the number of authors shown on this cover page is limited to 10 maximum.



Citation: Frånberg M, Gertow K, Hamsten A, PROCARDIS consortium, Lagergren J, Sennblad B (2015) Discovering Genetic Interactions in Large-Scale Association Studies by Stage-wise Likelihood Ratio Tests. PLoS Genet 11(9): e1005502. doi:10.1371/journal.pgen.1005502

Editor: Gregory S. Barsh, Stanford University School of Medicine, UNITED STATES

Received: February 25, 2015

Accepted: August 14, 2015

Published: September 24, 2015

Copyright: © 2015 Frånberg et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Because of restrictions based on privacy regulations and informed consent of the participants, data cannot be made freely available in a public repository. However, data can be obtained upon request. Requests for data should be directed to Hugh Watkins (<u>hugh.watkins@cardiov.ox.ac.uk</u>) or Anders Hamsten (<u>Anders.Hamsten@ki.se</u>) on behalf of the PROCARDIS consortium.

Funding: This study was funded by grants from the Swedish e-Science Research Centre and the Heartlung Foundation (20140433). BS is supported by the Magnus Bergvall foundation and the Foundation for **RESEARCH ARTICLE**

Discovering Genetic Interactions in Large-Scale Association Studies by Stage-wise Likelihood Ratio Tests

Mattias Frånberg^{1,2,3}*, Karl Gertow¹, Anders Hamsten¹, PROCARDIS consortium¹, Jens Lagergren⁴, Bengt Sennblad^{1,3}

 Atherosclerosis Research Unit, Department of Medicine, Solna, Karolinska Institutet, Stockholm, Sweden,
 Department of Numerical Analysis and Computer Science, Stockholm University, Stockholm, Sweden,
 Science for Life Laboratory, Stockholm, Sweden, 4 School of Computer Science and Communications, KTH Royal Institute of Technology, Science for Life Laboratory, Swedish e-Science Research Centre, Stockholm, Sweden

¶Membership of the PROCARDIS consortium is listed in the Acknowledgments.
* mattias.franberg@ki.se

Abstract

Despite the success of genome-wide association studies in medical genetics, the underlying genetics of many complex diseases remains enigmatic. One plausible reason for this could be the failure to account for the presence of genetic interactions in current analyses. Exhaustive investigations of interactions are typically infeasible because the vast number of possible interactions impose hard statistical and computational challenges. There is, therefore, a need for computationally efficient methods that build on models appropriately capturing interaction. We introduce a new methodology where we augment the interaction hypothesis with a set of simpler hypotheses that are tested, in order of their complexity, against a saturated alternative hypothesis representing interaction. This sequential testing provides an efficient way to reduce the number of non-interacting variant pairs before the final interaction test. We devise two different methods, one that relies on a priori estimated numbers of marginally associated variants to correct for multiple tests, and a second that does this adaptively. We show that our methodology in general has an improved statistical power in comparison to seven other methods, and, using the idea of closed testing, that it controls the family-wise error rate. We apply our methodology to genetic data from the PRO-CARDIS coronary artery disease case/control cohort and discover three distinct interactions. While analyses on simulated data suggest that the statistical power may suffice for an exhaustive search of all variant pairs in ideal cases, we explore strategies for a priori selecting subsets of variant pairs to test. Our new methodology facilitates identification of new disease-relevant interactions from existing and future genome-wide association data, which may involve genes with previously unknown association to the disease. Moreover, it enables construction of interaction networks that provide a systems biology view of complex diseases, serving as a basis for more comprehensive understanding of disease pathophysiology and its clinical consequences.



Old Servants (2014-00090). KG is supported by the Heart-lung Foundation. JL is supported by The Swedish Research Council (2013-4993). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

Author Summary

Many of our common diseases are driven by complex interactions between multiple genetic factors. Disease-specific, genome-wide association studies have been the prominent tool for cataloging such factors, by studying the genetic variation of a gene in a population and its association with the disease. However, these studies often fail to capture interactions between genes despite their importance. Interactions are notoriously difficult to investigate, because testing the large number of possible interactions using contemporary statistical methods requires very large sample sizes and computational resources. We have taken a step forward by developing a new statistical methodology that significantly reduces these requirements, making the study of interactions more feasible. We show that our methodology makes it possible to study interactions on a large scale without compromising the statistical accuracy. We further demonstrate the utility of our methodology on data relating to coronary artery disease and discover three distinct interactions that might provides new clues to the pathophysiology. The study of genetic interactions have the potential to link disease genes together into disease networks that provide a more detailed description of the interaction between genes and how it drives the disease.

Introduction

Cardiovascular disease, cancers, diabetes and chronic obstructive pulmonary disease, accounting for almost 60% of the causes of death 2013, globally [1], are all examples of complex diseases. A complex disease is characterized by an intricate system of interactions between genetic, epigenetic, other intrinsic factors, and environmental factors, that constitutes its pathophysiology. The genetic architecture of many common complex diseases is poorly understood. For example, the 46 robustly associated variants that have been found for coronary artery disease (CAD) only explain 10.6% of the estimated heritability; this was shown in a recent meta-analysis of almost 200,000 individuals [2]. The same pattern of unexplained, or missing, heritability is found in most common complex diseases [3]. Assuming that the estimated heritability is correct, the possible explanations for the high ratio of missing heritability include 1) a large number of causal genetic variants, each with a small effect, 2) sequence variation that is commonly excluded from analysis, e.g. copy number variation or rare variants, 3) other commonly unmeasured heritable components, e.g. heritable epigenetic modifications, and 4) interaction effects between common variants. Moreover, any combination of these explanations is plausible. In this paper we focus on the inference of interaction in genetic association studies; this is sometimes called epistasis, epistatic interaction or genetic interaction; here we will refer to it as genetic interaction or simply interaction.

Genetic interactions are characterized by two or more variants producing an unexpected phenotype that is not easily explained by the marginal effects of the individual variants. Extensive studies in model organisms have shown that genetic interactions are common phenomena [4]. The field was pioneered by Bateson [5], who studied genetic interactions in plants and chicken. More recently, synthetic lethal interactions (in which the simultaneous occurrence of two mutations, by themselves without effect, lead to cell death) have been studied extensively in yeast and *Caenorhabditis* [6–8] and interactions between quantitative trait loci have been studied in mouse, *Drosophila* and *Caenorhabditis* [9–11]. Since interactions are widespread in other organisms, it seems unlikely that such effects would not exist in humans. Furthermore, genes are linked in metabolic, regulatory and signaling pathways and it is likely that this will be reflected as interactions between variants, as has been shown for transcriptional regulation in

Drosophila [12]. Therefore, studies of genetic interactions have a strong potential to provide important insights about disease biology—specifically, interactions reflect dependencies in pathophysiology and may allow predictions of effects (and side effects) that are relevant for disease prognosis and treatment.

Several approaches have been developed to study genetic interactions (see [13-15] for three excellent methodology reviews). In medical genetics, the prevalent tool for modeling single variant association in unrelated individuals has been generalized linear models (GLM). The advantages of GLMs are flexibility in modeling the phenotype, easy interpretation and straightforward adjustment for confounders. Although the GLM framework can model both discrete and continuous outcomes; we will, in this work, concentrate on the case-control outcome. Studies of interactions are, however, not without issues. Firstly, the identification of interactions depends on the *scale* relating the genotypes to phenotypes. Secondly, because the GLMs are fitted by iterative procedures, the computational burden is high. Thirdly, straightforward multiple testing correction leads to low statistical power. We now elaborate briefly on these three issues.

The dependency of GLMs on a *scale* sometimes causes confusion [16, 17]. The scale is determined by a link function that maps the phenotype to the linear predictors. For example, for two predictors a and b, the phenotype y can be determined by an additive (y = a + b) or by a multiplicative $(y = e^{a+b})$ model. A commonly used link function in case/control studies is the logit, which is used in logistic regression. This displays a combination of mathematically favorable properties: it models the case/control selection bias, the parameters have minimal sufficient statistics, and it is the maximum entropy null model [18]. However, the choice of scale is to a large extent a modeling issue and should not be based on mathematical convenience alone. For example, when, for a set of variants, the presence of a risk allele in any single variant is sufficient to cause the disease, the log-complement link function yields an appropriate model [17, 19]. Ultimately, the best choice of scale depends on the unknown biological model that has generated the data. The choice of scale is very problematic because, even if the true model underlying the data displays interaction, it is often possible to select a scale that diminishes the interaction effect [20]. Conversely, if the true model does not display interaction, then there is another scale that, in the asymptotic case, will display interaction [17]. In response to this, Knol and VanderWeele [21] suggest that the p-value of an interaction should be reported on a set of reasonable scales to show whether the interaction seems invariant of scale. We follow this suggestion and, furthermore, extend it by constructing a test for interaction that is invariant over a set of link functions.

A different approach builds on the rationale that if, for interacting variants, certain combinations of alleles affect disease risk, then this would be reflected in differential enrichment for these allele combinations between cases and controls, and therefore a difference in their linkage coefficients (LD_{cases} and $LD_{controls}$). The LD-contrast test [22] compares the normalized difference of LD_{cases} and $LD_{controls}$ as a χ^2 -distributed statistic for interaction. A recently been proposed version of the LD-contrast test [23] uses genotypes recoded to a pair of binary variables (according to model of inheritance).

A third approach, the multi-factor dimensionality reduction (MDR) [24], uses dimensionality reduction techniques to recode the 3×3 penetrance matrix into a binary variable that optimally classifies cases and controls. This is then evaluated by cross-validation and a permutation procedure is used to estimate significance. Several variants of MDR have been developed [25, 26].

One common approach to improve the computational complexity, has been to introduce the naive assumption that it is impossible for two variants to be simultaneously associated with the phenotype unless they interact. Examples include the method of [27], using a log-linear GLM, and many of the variations of the MDR method, including the original one [24, 25].

Under this naive assumption it is, in the GLM setting, sufficient to compare three models: two single variant association models and the saturated model, which will represent interaction. The parameters of these models can be efficiently estimated since they all have closed form solutions. Unfortunately, this simplification allows interactions to be incorrectly inferred between two variants that both are associated with a main effect, but there is no interaction (we will refer to this as *double main* association). As a consequence, genuine genetic interactions may be obscured by these *double main* associations [28]. In this work, we will focus on inference of genuine interactions.

Finally, the reduction in statistical power implied by correction for multiple tests constitutes a major limitation for performing interaction studies on a larger scale. For an investigation of interaction between all pairs of a set of n = 500,000 variants, a Bonferroni correction for n(n-1)/2tests gives a significance threshold of $4.0 \cdot 10^{-13}$, which is considerably lower than the corresponding significance level $1.0 \cdot 10^{-7}$ for a standard single variant analysis. The burden of multiple tests grows exponentially with the number variants involved in the tested interactions, and, henceforth, we will limit ourselves to the case of pair-wise interactions. Various screening strategies have been applied in attempts to improve power. These may use prior information that identify a smaller set of candidate variant pairs (we will investigate two such approaches in our analysis of biological data) or they may be based on the data at hand. An example of the latter include the screening test of Marchini [29], which removes variant pairs lacking a marginal effect for one or both of the participating variants. Millstein et al. [30], using a reasoning similar to the LD-contrast test above, suggested a LD_{cases} screening test for significant linkage enrichment among cases. However, observing that this induced a bias in the subsequent main test, they also proposed a LD_{cohort} screening test. The latter test relies on the linkage enrichment in cases also showing as a linkage enrichment of the pooled cases and controls, but formally does not use any prior information about disease state. Various combinations of screening and main tests have been proposed: marginal screen with logistic GLM main test [29], LD_{cohort} screen with logistic GLM main test [30, 31], and LD_{cases} screen with LD-contrast main test [23].

In this work we introduce a stage-wise multiple testing methodology that exhaustively tests all variant pairs. In this methodology, a sequence of hypotheses is considered in order of increasing complexity. Only variant pairs that cannot be explained better by a simpler hypothesis compared to the most complex hypothesis (representing interaction) are tested at subsequent stages. This is conceptually different from the screening approach by Marchini [29], which instead requires that a variant pair fits an intermediate screening hypothesis (of single marginal association) better than the simplest hypothesis (of no association) for it to be tested at the subsequent stage. Because the hypotheses considered are closed under intersection, we show, in two situations, that the family-wise error rate is controlled. Furthermore, since the models under the simpler hypotheses can be estimated efficiently, our methodology allows the use of full GLMs. The multiple testing correction is alleviated and results in a substantial increase of power compared to the Bonferroni correction. We also construct a scale-invariant test for interaction using several link functions. Furthermore, we assess a set of statistical methods for inferring genetic interactions on synthetic data and show that our methodology improves on these. Lastly, we discover three distinct interactions that are associated with CAD, of which one includes a novel locus.

Results

Theory

In this section, we describe our multiple testing methodology, which is aimed at large-scale pairwise interaction testing. We show that it gains additional power by separating a complex

hypothesis into stages of simpler null models. We have derived two methods that rely on different assumptions, having different effects on the bounds of the family-wise error rate (FWER). We start by briefly reviewing general linear models (GLM), which we use to express our model of interaction, as well as the simple null models. Frequently, when GLMs are applied in pairwise interaction testing, FWER is bounded using Bonferroni.

Interaction in the generalized linear model. In the framework of generalized linear models (GLM), the penetrance p_{ab} of two given genotypes $X_1 = a$ and $X_2 = b$, where $a, b \in \{0, 1, 2\}$, is modeled by

$$g(p_{ab}) = \alpha + \beta_a + \gamma_b + \delta_{ab}$$

where β_a and γ_b represents the *main effect* on the phenotype of *a* and *b*, respectively, δ_{ab} represents the *interaction effect* of *a* and *b*, and *g* is the *link function*. The link function is invertible and determines the scale. Sometimes, this model is restricted by adding constraints forcing an additive allele effect, that is, $\beta_2 = 2\beta_1$, $\gamma_2 = 2\gamma_1$, $\delta_{21} = \delta_{12} = 2\delta_{11}$, and $\delta_{22} = 4\delta_{11}$. However, in the present methods we will *not* assume an additive allele effect. Of note, to avoid over-parameterization, we set $\beta_0 = \gamma_0 = \delta_{00} = \delta_{01} = \delta_{02} = \delta_{10} = \delta_{20} = 0$.

A standard likelihood ratio test of interaction can be obtained by comparing the likelihood of this full model to that of a null model in which $\delta_{ab} = 0$ for all a, b. In our stage-wise methodology, we will use such tests for various null models, corresponding to subsets of the parameters. The degrees of freedom are determined by the difference in the number of free parameters; for example, if we want to test whether β_a , γ_b or δ_{ab} is different from 0 (i.e., the reduced model is $g(p_{ab}) = \alpha$), we would compare against a χ^2 distribution with 9 - 1 = 8 degrees of freedom.

Stage-wise testing for interactions. We now describe how to apply our GLM to a stage-wise test of interaction to a set *S* of multiple variant pairs. We test the interaction hypothesis by a stage-wise comparison of the full model

• $H_A: g(p_{ab}) = \alpha + \beta_a + \gamma_b + \delta_{ab}$

to the following null models

- $H_1: g(p_{ab}) = \alpha$
- $H_2: g(p_{ab}) = \alpha + \beta_a$
- $H_3: g(p_{ab}) = \alpha + \gamma_b$
- $H_4: g(p_{ab}) = \alpha + \beta_a + \gamma_b.$

The H_1 null model represents no association, H_2 and H_3 represent the cases where one single variant is associated (we will refer to this as *single main* association) and H_4 represent the case where both variants simultaneously have a main effect association without interaction (*double main* association). The tests are performed sequentially. In the first stage, H_1 is tested against H_A for all variant pairs, but subsequent stages only consider variant pairs for which the null models in previous stages have been rejected. This allows us to vary the multiple testing correction across stages, and thereby improve power compared to Bonferroni correction, as shown, below, in Results section Statistical power to detect interactions. The exact details of this correction constitute the difference between our two methods.

In the first method, the *static* method, we assume that the exact number of variant pairs *belonging* to each stage is known. Intuitively, a variant pair belongs to a stage if the null model at this stage is the simplest model that is correct for the pair. To preserve the FWER, we introduce weights { w_s , $s \in [4] : \sum_{s \in [4]} w_s = 1$ }, one for each stage, that adjust the p-value thresholds



Fig 1. Illustration of the rejection procedure. a) We have a set of four hypotheses that are closed under intersection. We start at stage 1 by testing the simplest null hypothesis H_1 for each variant pair; the p-value threshold α for this test is corrected for the total number of pairs. In the figure H_1 is accepted in the first and last pair, and these pairs will not be tested in the subsequent stages. We then continue through the null hypotheses from simple to complex but correcting the α for each stage only for the expected number of pairs, in the *static* method, or the actual number of pairs, in the *adaptive* method, that are tested at this stage. Finally if all null hypotheses could be rejected for a specific pair *i* (e.g., snp2 and snp3 in the figure), we declare pair *i* to be interacting. b) Pseudocode describing the *static* stage-wise testing method. Variant pairs are ordered from 1 to *n*. Null hypotheses are ordered from 1 to *m* in any order that respects the partial order of how they are nested. Only pairs for which the null hypothesis was rejected in the previous step are considered in the current step. The p-value for testing null hypothesis *j* for pair *i* is p_{ij} . Rejected hypotheses for all *m* tests could be rejected. c) Pseudocode describing the *adaptive* method. The hypotheses of interaction for pair *i* is accepted only if the null hypotheses for all *m* tests could be rejected. c) Pseudocode describing the *adaptive* method. The overall algorithm is the same as the *static*. However, the significance threshold is now determined by the total number of rejections in the previous stage $|R_{j-1}|$.

doi:10.1371/journal.pgen.1005502.g001

for the four stages of tests. Let K_s be the number of variant pairs belonging to a stage $t \ge s$, and p_{is} be the p-value of stage *s* for pair *i*. If $p_{is} < w_s \alpha/K_s$, the pair *i* is tested for stage s + 1. The idea is illustrated in Fig 1a and the algorithm is outlined in Fig 1b. A generalized version of the closed testing principle [32] can be used to show that this method controls the FWER, a proof is provided in S1 Text. The adjusted p-value is defined [33] and can be computed by

$$\tilde{p}_i = \max_s \quad \frac{K_s p_{is}}{w_s}.$$

The following is an example on how to estimate the number of hypotheses in each stage. Let N be the number of genotyped variants and M be the number of marginally associated variants (which, e.g., can be taken from a meta analysis). Then estimates of the *static* multiple testing correction, K_{s} , for each stage are, in order, $\frac{N(N-1)}{2}$, $N \cdot M$, $N \cdot M$ and $\frac{M(M-1)}{2}$.

The second method, the *adaptive* method, only corrects for the number of rejections in the previous stage. This more flexible and powerful method is tailored for situations where an estimate of the number of marginally associated variants is not available. Because the multiple testing corrections become dynamic, the closed testing theorem cannot be directly applied. We therefore need to make the additional assumption that alternative hypotheses are asymptotically rejected, which is true for any consistent test. A proof is provided in <u>S1 Text</u>.

The *adaptive* method is identical to the *static* method, with the exception that K_s is replaced by $|R_{s-1}|$ where R_s is the set of rejected hypotheses in stage *s*, and R_0 is the total number of pairs. The algorithm is outlined in Fig 1c. A conservative estimate of the adjusted p-value is

$$\tilde{p}_i = \max_s \quad \frac{|R_{s-1}|p_{is}}{w_s}$$

since as you alter the significance threshold for a stage, different number of hypotheses, |R|, will be present on the following stages.

An advantage with our methodology is that H_1 to H_3 can be efficiently computed [27]. The tests for H_1 to H_3 is also invariant of the scale as shown in <u>S1 Text</u>. Moreover, key for the improved power of our methodology is two assumptions relating to the test for H_1 : (1) that it has high power to identify interactions, and (2) that it, at the same time, is effective in removing false positives and thereby reducing the severity of multiple testing in the following stages. In our simulation results assessing assumption (1), the H_1 test is more powerful than that for H_4 , despite the increase in degrees of freedom (cf. Results section Statistical power to detect interactions). Moreover, for complex diseases, the number of associated variants identified in large meta analyses is commonly less than 100 [3], suggesting that assumption (2) will not be a bottleneck. We also investigate the practical performance of the methods.

Scale-invariance test. The choice of link function, *g*, has a substantial influence on any study of interactions based on GLMs. Ideally, we want the inferred interaction to be invariant to the scale, that is, the rejection the null model (of no interaction) should hold regardless of the link function used [21]. We will now describe a scale-invariant test that formalizes this notion.

With slight abuse of notation, we replace hypothesis H_4 in the previous section with

•
$$H'_4: \bigcup_{g\in G} \{g(p_{ij}) = \alpha + \beta_i + \gamma_j\},\$$

where *G* is some set of pre-selected link functions (note that the models H_1 , H_2 and H_3 are saturated and changing the link function will not have an effect). The test for H'_4 is constructed according to the intersection-union principle [34], that is, if q_s is the p-value for scale *s* then the combined p-value *q* is

$$q = \max q_s$$

that is, the test requires significance for all link functions; we will refer to this test as the *scale invariance test*. Notice that the choice of the set *G* of link functions will depend on the study. For example, *G* may contain a single link function if there is strong prior evidence for this link function. Furthermore, the study design will influence the choices of *G*; for example, an odds ratio-based link function is unaffected by the sampling-bias introduced in case-control studies and, omission of a such a link function may lead to the wrong conclusion. Here, we investigate the impact of a set of commonly used link functions, shown in <u>Table 1</u>.

Statistical power to detect interactions

In this section, we will give an account of three investigations of statistical power that all indicate the utility of our stage-wise methodology. The generation of simulated data used in these investigations are described in Material and methods section Generation of synthetic data for estimation of statistical power.

Table 1. Link functions	s, for a GLM with a binar	y outcome, used	l in this study.
-------------------------	---------------------------	-----------------	------------------

Link	g(p _{ij})
Additive penetrance (identity)	p _{ij}
Multiplicative penetrance (log)	log(p _{ij})
Genetic heterogeneity (log-complement)	$\log(1 - p_{ij})$
Additive odds (odds)	$\frac{\rho_{ij}}{1-\rho_{ij}}$
Multiplicative odds (logit)	$\log\left((rac{m{ ho}_{ij}}{1-m{ ho}_{ij}} ight))$

doi:10.1371/journal.pgen.1005502.t001



Fig 2. The power of the first and last test under a double dominant interaction model. The x-axis is the heritability of the model. The y-axis is the statistical power. The colored lines correspond to two different tests: the one performed in the first stage that tests the null hypothesis of no interaction, H_1 , (red), and the one performed in the last stage that specifically tests the interaction parameters, H_2 , (blue). The logit link function and a nominal significance level of 0.05 was used for the analysis.

doi:10.1371/journal.pgen.1005502.g002

GENETICS

PLOS

The intuitive idea behind the stage-wise methodology is that we aim to (1) reduce the number of tests in later stages compared to earlier, while (2) asserting that actual interactions advance to later stages. We show in the Results section Analysis of biological data, below, that the number of tests in the last stage is in fact substantially reduced, suggesting that aim (1) is unlikely to be a problem. Here, we have investigated aim (2 by comparing the power of the tests in the first and last stage. That is, for data generated from H_A , we compare the power of the likelihood ratio test of H_1 against H_A to that of the test of H_4 against H_A . Indeed, the results in Fig.2 (using data generated from a double dominant interaction model) suggests that the test in the first stage, at least under these conditions, have substantially greater power than that in the last stage. However, the test in the first stage can obviously not be used as a test for interaction by itself, since it measures any kind of association to the phenotype, including, for example, pairs for which only one of the variants is associated.

We further investigated the distribution of statistical power of seven methods using simulated data generated from the spectrum of all possible interaction models, following the ideas of [35] (see <u>Material and methods</u> section Generation of synthetic data for estimation of statistical power for details). The first of these methods is our *static* method, and the remaining methods include four methods based on a logit-link GLM with different screening strategies, *Logistic* (without screening), *Marginal+logistic* [29], *CSS+logistic* [30] and R²+*logistic* [31]) and two methods based on the *LD*-contrast test with different screening strategies, *LD-contrast* (without screening), and *Sixpac* [23] (a $LD_{cases}+LD$ -contrast method), for details, see <u>Material</u> and methods section Comparison of statistical methods. It should be noted that none of the latter six methods are scale-invariant—one may expect that this property would enhance their



Fig 3. The exceedence distribution of power over all possible interaction models with a specific heritability. For each plot, the *x*-axis shows a threshold, *t*, for power to detect an interaction among 10^{12} variant pairs, and the corresponding y-axis shows the fraction of models that have a power greater than or equal to *t*. The rows correspond to the sample size of a balanced design e.g. 2000 indicates 2000 cases and 2000 controls. The columns correspond to the heritability of the models. Six methods for inference of interactions *Logistic*, *Marginal+logistic*, *CSS+marginal*, R^2 +marginal, *LD*-contrast, and *Sixpac*; see text for details), are compared to our *static* stage-wise scale-invariant method. The line colors used to denote the different methods are shown in the legend to the right.

doi:10.1371/journal.pgen.1005502.g003

PLOS GENETICS

power. For simplicity of simulations, we only evaluated the *static* method here; however, since the *adaptive* method is more powerful than the *static*, this can also be viewed as a conservative estimate of the power of the *adaptive* method. As can be seen in Fig 3, the *static* method consistently has greater power than the other approaches. The *marginal+logistic* method performs best of the remaining methods, while the the LD-*contrast* method have the worst performance. In S1–S4 Figs, we also report the result of a more computationally intensive power comparison, including the above methods, as well as our *adaptive* stage-wise scale-invariance method and the Model-based MDR (*MB-MDR*) method [26] (see S2 Text for details). These results corroborate those above, that is, for most models our stage-wise methods performs better than the other methods (see further discussion in S2 Text).

Intuitively, when more variants are associated with the phenotype in our stage-wise methodology, the multiple testing correction in the intermediate stages becomes larger, and therefore statistical power is reduced. For this reason, we investigated how the statistical power depends on the number of associated variants using data simulated from the double-dominant interaction model (see <u>Material and methods</u> section Generation of synthetic data for estimation of statistical power). As shown in <u>Fig 4</u>, the power decreases as the number of associated



Fig 4. The dependence of power on the estimated number of associated variants. The x-axis is the heritability and the y-axis is the estimated power. The colored dashed lines correspond to our stage-wise test using different number of associated variants, as indicated by the legend to the right. The black solid line corresponds to the logistic regression method using Bonferroni correction (which does not depend on the estimated number of associated variants). The power estimates are based on data simulated from the double dominant interaction model.

doi:10.1371/journal.pgen.1005502.g004

PLOS

GENETICS

variants increases. Because of the additional penalty of the weight, the *static* method can have lower power than directly testing interaction using a Bonferroni correction, precisely when M $(M - 1) > w_4 N(N - 1)$ (where N is the total number of variants and M is the number of associated variants). It can be noted that for our biological data, $M(M - 1) = 306 \ll w_4 N(N - 1) =$ 346,035,421.8 (based on the N = 33,963 tested variants and the M = 18 robustly associated CAD variants present on the IBC-chip, cf. <u>S1 Table</u>).

Family-wise error rate on the class of null models

Both the *static* and *adaptive* stage-wise methods are based on the likelihood ratio test, which is asymptotically correct. As we show in <u>S1 Text</u>, the *adaptive* method controls the FWER asymptotically. For the *static* method on the other hand, we can even show that the FWER is controlled for any data size. Consequently, it is interesting to investigate the behavior of both methods on finite data and compare it with that of the same seven methods as in the power comparison (*Logistic*, *Marginal+logistic*, *CSS+logistic*, R²+*logistic*, LD-*contrast*, *Sixpac*, and *MB-MDR*). We considered two cases, one close to the assumptions of our methods, and one designed to be challenging. We investigated these two cases using simulated data (see further <u>Material and methods</u> section Generation of synthetic data for estimation of statistical power).

In the first case, we used each of our null models, H_1 : no association, H_2/H_3 : *single main* association, and five models for *double main* association (H_4) with the identity, log, log-complement, odds and a logit link functions, respectively, to generate the phenotype based on a single pair of variants. The first seven rows in Table 2 show that both the *static* and *adaptive*



Table 2. Estimated family-wise error rate for all methods under different null models, using a sign	ificance threshold of 0.05.
-----------------------------------------------------------------------------------------------------	-----------------------------

Model ¹	Adaptive	Static	Logistic	Marginal+logistic	CSS+logistic	R ² +logistic	LD-contrast	Sixpac	MB-MDR
No association	0.005	0.005	0.005	0.015	0.02	0.015	0.045	0.005	0.015
Single ²	0	0	0.02	0.015	0.025	0.05	0.045	0.005	0.02
Double ³ :identity	0.02	0.025	0.275	0.355	0.145	0.075	0.08	0.22	0.31
Double:log	0.01	0.01	0.025	0.035	0.035	0.035	0.03	0.01	0.035
Double:odds	0.035	0.02	0.955	0.975	0.54	0.16	0.66	0.985	0.99
Double:logit	0.015	0.015	0.04	0.03	0.035	0.07	0.035	0.005	0.035
Double:log-compl	0.03	0.025	0.03	0.045	0.035	0.035	0.03	0.01	0.04
Multi ⁴ :10	0.15	0.015	0.035	0.03	0.025	0.05	0.055	0.015	0.06
Multi:20	0.145	0.015	0.025	0.05	0.025	0.07	0.055	0.005	0.055
Multi:30	0.135	0.035	0.01	0.03	0.03	0.025	0.045	0	0.015

¹ Model from which data was generated

² Single main association

³ Double main association

⁴ Multivariate additive GLM

doi:10.1371/journal.pgen.1005502.t002

methods accurately control the FWER under these circumstances (i.e., FWER is below or close to the expected value of 0.05). All seven other methods control the FWER for the no association and *single main* models. However, for the *double main* models, they control FWER only on the multiplicative scale (i.e., with the log, logit and log complement link functions). For the remaining models (*double main*:identity and *double main*:odds) these methods fail to control FWER, with the exception of the R^2 +*logistic* that controls FWER for the *double main*:identity model.

In the second case, where we attempted to construct instances that challenge the additional asymptotic assumptions made in the *adaptive* method. The phenotype was here determined by an multivariate additive GLM with logit link function on a set of $L \in \{10, 20, 30\}$ markers. The parameter distributions were chosen with the intention to let only a small and difficult subset of the variant pairs to reach the stage they belong to. The last three rows in <u>Table 2</u> show that the *static* method controls the FWER, but suggest that for the *adaptive* method, FWER is inflated by approximately a factor 3 compared to the desired rate. The remaining methods controls FWER in this setting, possibly an effect of the data being generated on a multiplicative scale.

Analysis of biological data

We applied our stage-wise methodology on genome-wide CAD case-control data from the PROCARDIS study, using our five link functions. To enhance such a large-scale analysis, we explore three strategies for selecting subsets of variant pairs to test. The first strategy represents a genome-wide approach, while the latter two strategies were designed to a priori enrich for pairs likely to exhibit interaction. For the same reason, our main focus will be on the more powerful adaptive method, combined with validation of any significant discoveries in a separate cohort.

In the first strategy all 229,050,992 pairs, for which the product of the minor allele frequencies > 0.04, were selected. The stage-wise methodology subsequently reduced the number of pairs to 15269, 7712 and 93. This analysis resulted in seven variant pairs that were significant for at least one link function, see <u>Table 3</u>. We used genomic proximity to coarsely estimate genes corresponding to these variant pairs. One variant pair, indicating an interaction between IL1R1 and CDNK2B-AS1, was significant on the additive odds scale, only. The p-

All vs all strate	egy			Oc	lds		Penetrance	
SNP 1 ¹	SNP 2 ¹	Locus 1 ²	Locus 2 ²	Add ^{3,4}	Mul ^{3,5}	Add ^{3,6}	Mul ^{3,7}	Het ^{3,8}
rs3917245	rs1412832	IL1R1	CDKN2B-AS1	0.03746	0.1463	0.1412	0.2855	0.07222
rs4846770	rs518394	MIA3	CDKN2B-AS1	0.06048	0.03391	0.03419	0.0299	0.0468
rs17163313	rs518394	MIA3	CDKN2B-AS1	0.08093	0.04386	0.04438	0.03136	0.0617
rs17163313	rs2069418	MIA3	CDKN2B-AS1	0.1422	0.06734	0.06773	0.04921	0.1015
rs2378584	rs518394	MIA3	CDKN2B-AS1	0.08615	0.04893	0.04929	0.04398	0.06707
rs4846770	rs2069418	MIA3	CDKN2B-AS1	0.09625	0.04741	0.04753	0.04296	0.06983
rs17163301	rs518394	MIA3	CDKN2B-AS1	0.08487	0.04869	0.04899	0.0457	0.06637
CAD loci vs al	l strategy			Oc	lds		Penetrance	
SNP 1	SNP 2	Locus 1	Locus 2	Add	Mul	Add	Mul	Het
rs17465637	rs518394	MIA3	CDKN2B-AS1	0.08232	0.04734	0.04769	0.03447	0.06405
HumanNet strategy			Oc	lds		Penetrance		
SNP 1	SNP 2	Locus 1	Locus 2	Add	Mul	Add	Mul	Het
rs4694178	rs583104	IL8,CXCL6	PSRC1	0.04876	0.04876	0.04876	0.04876	0.04876
rs4694178	rs602633	IL8,CXCL6	PSRC1	0.006329	0.006329	0.006329	0.006329	0.006329

Table 3. Significant pairs using different strategies.

¹Variant pairs that are statistically significant on at least one scale.

²The loci most close, in genomic proximity, to the corresponding variant.

³Adjusted p-values; significant p-values are indicated by bold face; significance on all scales is required to pass the scale-invariance test.

⁴Additive odds ratio

⁵Multiplicative odds ratio

⁶Additive penetrance

⁷Multiplicative penetrance

⁸Genetic heterogeneity

doi:10.1371/journal.pgen.1005502.t003

values for the other scales were quite far from significance, indicating that this association is not scale-invariant. In other words, this interaction should be interpreted with care, as we cannot exclude the possibility that this is the effect of *double main* association, e.g., on the logit scale, without interaction. The remaining six variant pairs indicated an interaction between the genes MIA3 and CDNK2B-AS1. None of these passed the scale-invariance test (i.e., was significant for all link functions). However, for most of these variant pairs, the p-values for all scales are of the same magnitude and reasonably close to the significance level of 0.05 (see, e.g., the rs4846770–rs518394 variant pair), perhaps suggesting that this could be an effect of insufficient power rather than scale dependency.

In the second strategy 314,445 pairs were selected based on loci previously associated with CAD. This is based on the common hypothesis that some robust CAD associations may be the marginal effects of interacting variant pairs. Candidate pairs were formed by taking each of the previously associated CAD variants, see <u>S1 Table</u>, and combining it with each other variant. Interestingly, similar to the results from the all-vs-all strategy above, one variant pair indicating an interaction between MIA3 and CDNK2B-AS1 was significant for several link functions, but again, just, failed the scale-invariance test, see <u>Table 3</u>. Somewhat unexpectedly, this variant pair did not coincide with any of those in the all-vs-all analysis. However, it turns out that, while variants for both MIA3 and CDNK2B-AS1 have previously been robustly associated to CAD (see <u>S1 Table</u>), these variants did not include any member of the top-scoring variant pairs in the all-vs-all analysis. This enrichment strategy might therefore have been suboptimal.

In the third strategy, we used prior information from HumanNet [<u>36</u>], a probabilistic functional gene network that links genes for which significant evidence of interaction have been provided in one or more omics experiments; this resulted in 2,319,906 variant pairs. We found two variant pairs that were significant for all five link functions, thereby passing the scale-invariance test, see <u>Table 3</u>. For each of these two pairs, genomic proximity suggests that an interaction between PSRC1 and CXCL6 is associated to CAD, and, thus, may play a role in its pathophysiology. The exact mechanism of the interaction is, however, unknown, and the evidence for it in HumanNet was merely reported as co-expression between human genes.

The maximal effect size for each discovered interaction range from 0.4718 to 1.379 (S2 Table); as a reference, the effect sizes for robustly CAD-associated single variants are commonly around 0.285 [37]. While, after adjustment for age, sex, smoking, and population stratification, most effect sizes were reduced, this was not the case for the CXCL6-PSRC1-related interactions (see S3 Table). The penetrance pattern of one of the CXCL6-PSRC1 variant pairs, rs4694178 and rs602633, is shown in S5 Fig. Of note, it shows a marked directional change in risk for individuals carrying the major rs4694178 homozygote and the minor rs602633 homozygote.

We then investigated the reproducibility of the CXCL6-PSRC1-related interaction on a non-overlapping sub-cohort of PROCARDIS. This sub-cohort consists of 1797 cases and 2677 controls, which were genotyped on the Illumina Human1M Quad chip. The exact variants of the significant pair were not genotyped, and was therefore imputed (and hard-called) using the 1000 Genomes phase 3 reference panel. We tested interaction directly using a GLM combined with our link functions. This resulted in the p-values 0.174, 0.241, 0.103, 0.056, 0.156, for the identity, log, log-complement, odds and logit link functions respectively. We note that, while the p-value for the odds scale is close to significance, the replication clearly did not pass the scale-invariance test. Despite this, the penetrance patterns for different allele combinations were very consistent between discovery and replication analyses, compare <u>\$5</u> and <u>\$6</u> Figs. Of note, is that the minor allele frequency for rs602633 is relatively different between the two cohorts, see <u>S4 Table</u>. We, furthermore, expanded the search to the ten closest variants on both sides of both significant variants. The best variant pair, rs11730560 and rs1277930, reached nominal significance, and the p-values were 0.023, 0.0311, 0.014, 0.0084, 0.0209, again for the identity, log, log-complement, odds and logit link functions respectively. It did, however, not pass multiple testing accounting for all the 380 tested variant pairs.

We also performed analyses using the *static* method assuming 100 marginally associated variants with the same search strategies, but no variant pair was significant on any scale for any of the strategies. This may be a consequence of the expected lower power of the *static* method.

Discussion

We have introduced a new stage-wise methodology that is statistically and computationally efficient for large-scale inference of genetic interactions. We have derived two separate methods: The first is the *static* method that uses *a priori* estimated multiple testing correction factors; here we have used the number of published robustly associated CAD SNPs to obtain such an estimate. The second *adaptive* method does not rely on the assumption of known correction factors, but uses the number of associated variant pairs at each stage to compute the multiple testing factors. To the best of our knowledge, this is the first method that uses the idea of a closed set of hypotheses to perform an exhaustive pairwise scan of interactions. We have shown that this stage-wise method performs better on a large number of directly testing all possible variant pairs for interaction, we use a sequence of more general association tests as a filter to reduce the number of pairs until only potential interactions remains. This shifts much of the multiple testing burden from the final interaction test to the preceding general tests. Because

the tests leading up to the interaction test in general are more powerful (i.e., interactions will not be discarded), this results in higher overall power.

Our simulation results show that our new methods in general have higher statistical power than other common interaction inference methods. For certain specific models and low MAFs, the *Sixpac* method [23] perform relatively well, but its performance over the spectrum of all possible interaction models is low. The simulations suggest that, in ideal cases, it may be possible to infer interactions using our stage-wise methodology even when correcting for 10^{12} pairs, since each stage greatly reduces the number of tested interactions. However, we conjecture that, in practice, it will be important to take advantage of prior information in order to reduce the number of tested interactions; for example, we used information from the HumanNet database to select candidate interactions. Moreover, the methodology presented in this paper can also be combined with screening procedures such as LD_{cohort} [30, 31] or the efficient *probable approximate complete search* algorithm of [23]. This may give even further gains in power and computational speed.

Deciding which scale to work on (i.e., which link function to use, see <u>Table 1</u>) can be troublesome and many researchers advocate a favorite scale for statistical or biological reasons. Testing on a single scale will improve the statistical power for interactions that fit that scale compared to testing multiple scales. However, if pairs of variants are additive on another scale, this approach will lead to an increased number of false positives, in the sense that there exists simpler models that explain the data. In our framework we offer a compromise: we display all pairs that are significant on at least one scale, but also provide a test that require significance on multiple scales. In this way, a researcher can interpret the significance of an individual scale in the context of the other scales. From our analyses of biological data, no particular scale appear to consistently be the critical one for the scale-invariance test.

We note that the scale-invariance test provides an advantage in terms of FWER control. While most other methods failed to control FWER for data generated with a link function that was sufficiently dissimilar from that underlying the method, the scale-invariance test allowed our methods to control FWER for data generated with any tested link function. Although the *static* method could be derived using closed testing, the derivation of the *adaptive* method relied on additional assumptions that may be difficult to satisfy in practice. We observed that this could cause inflation of the FWER under a specifically designed additive model with multiple weakly associated variants. We note that, while analytically straight-forward to work with, the FWER is known to be a conservative control of the experimental error at the expense of power [<u>38</u>]. One future direction could therefore be to investigate other error control measures, for example the false discovery rate (FDR) [<u>38</u>]. Moreover, there are several cases where the advantages, in terms of computational efficiency and statistical power, of the *adaptive* method may compensate for a relatively modest inflation in the FWER. Specifically, as validation is conventionally required in genetics studies, the *adaptive* method can be used as a powerful tool in the discovery phase of large-scale studies.

Our biological analysis identifies the well known CDKN2B-AS1 locus, or ANRIL, which encodes an anti-sense RNA [39]. The region contains several variants that are robustly associated with CAD but the pathophysiology of ANRIL is unknown. Interestingly, we detect an interaction between CDKN2B-AS1 and MIA3, another established CAD locus [2], potentially indicating a new lead on CAD pathophysiology. Variants in the CELSR2-PSRC1-SORT1 gene cluster have previously been shown to be associated to CAD and lipid traits [2], although the exact causal relation of the genes is unclear. Our results suggest that HumanNet's co-expression-based connection between CXCL6 and PSRC1 in fact mirrors a genetic interaction in CAD, supporting a role of PSRC1 in CAD (in line with recent results [40]). Moreover, inflammation has long been seen as an important component of the pathology of atherosclerosis, but

few inflammation genes have been implicated by genome-wide association studies [41], and only in meta-analyses. It is therefore interesting that in the two sets of variant pairs unbiased with respect to CAD, we find interactions involving genes clearly implicated in regulation of inflammation, i.e., the interleukin- and chemokine-related genes IL1R1 and CXCL6 (IL8). Of course, follow-up functional investigations are required to fully understand the potential pathophysiological consequences of these interactions.

Complex diseases are multi-gene and multi-factorial diseases characterized by complex interactions between genetic, regulatory, metabolic and environmental factors. The majority of complex disease genome-wide association studies have employed traditional association analyses of single genetic markers, which only have been able to explain a small fraction of the disease heritability. A perhaps more conclusive approach would be to reconstruct the complex dependencies between factors as an interaction network reflecting the disease pathophysiology. This approach, however, has so far been hampered by the lack of efficient methods for inference of interactions associated to disease. The *static* and *adaptive* methods are two effective ways to discover genetic interactions, and the flexibility of GLMs allows them to be applied to a wide range of different phenotypes. Genetic interactions, and in particular the construction of interaction networks explaining the pathophysiology of the disease, have a potential for clinical relevance, both in terms of prognosis, treatment and drug development. The ideas of stagewise testing is furthermore applicable outside medical genetics, whenever a large number of complex hypotheses are tested.

Materials and Methods

Ethics statement

The PROCARDIS study was carried out in accordance with the Helsinki Declaration and approved by the Institutional Review Board (IRB) at each one of the 4 recruiting centers: the Regional Ethics Review Board at Karolinska Institutet, Stockholm in Sweden (approval number 98-482 and 03-491), the IRB at the University of Munster, Munster, in Germany, the IRB at the Mario Negri Institute, Milano in Italy and the IRB at the University of Oxford, Oxford, United Kingdom. All study participants provided their written informed consent to participate in the study, a procedure approved by each one of the local ethical committees.

Biological data—the PROCARDIS cohort

A subset of the PROCARDIS cohort has previously [42] been genotyped with the Illumina IBC chip, a iSelect Custom Genotyping BeadChip [43]. This chip contains 48,742 variants in approximately 2,100 candidate genes that are believed to be involved in vascular disease processes. The subset of PROCARDIS used in this study are 3,162 cases and 3,353 controls of which 3,865 are males and 2,650 are females. The disease phenotype is CAD (including myocardial infarction). Multidimensional factor analysis indicated no significant population structure.

The following quality control was performed. We removed variants with a minor allele frequency < 0.05, with significant deviation from Hardy-Weinberg equilibrium $p < 10^{-6}$, and removed the variants from the X-chromosome to avoid confounding with gender, leaving us with 33,963 variants.

We performed simulations of all possible weight combinations with a precision of 0.1, the results can be seen in <u>S5 Table</u>. The choice seems to have little impact, and the best weight combination was 0.1, 0.3, 0.3 and 0.3, which is the one we used on biological data.

Strategies for the selection of variant pairs to test. We have considered three *a priori* determined strategies for selecting sets of variant pairs to investigate for genetic interaction.

Firstly, we considered all possible pairs of variants. We removed pairs for which the product of the minor allele frequencies was less than 0.04, as the likelihood ratio approximations become unstable for small minor allele frequencies. We also removed pairs in which the variants were less than 1 Mbp from each other, since the GLM models fail in presence of LD. Secondly, we considered pairs that contain at least one previously robustly associated variant. These robustly associated variants were taken from 46 published CAD-associated variants found in large-scale meta analyses [2], 18 of these were available on the IBC chip and can be found in <u>S1 Table</u>. Thirdly, we considered variant pairs whose corresponding genes have an edge in the HumanNet interaction database [<u>36</u>]. We downloaded HumanNet version 1.0 from the HumanNet web page. The genes in HumanNet is indexed by UniProt Ids and we therefore associated UniProt IDs to variants, as follows: Variants were annotated with ANNOVAR [<u>44</u>] and mapped to their nearest genes by genomic proximity; if there were more than one then the variant was mapped to both. We downloaded Uniprot IDs for all genes that were downloaded from Ensembl Biomart [<u>45</u>]. We then combined this information to assign each variant to one or more Uniprot IDs. A variant pair was then formed if the corresponding genes had an edge in HumanNet.

Generation of synthetic data for estimation of statistical power

We used two different simulation strategies for the power estimation. The first of these was used to compare the stage-wise scale-invariance method to other methods, see further next section. Models were constructed by enumerating all possible penetrance matrices displaying interaction for a single variant pair [35], as follows: The models were initially restricted to complete penetrance, that is, the penetrance is either 0 or 1, which allowed us to enumerate all $2^9 =$ 512 penetrance matrices. Only models considered to interact were included, here a model was defined as an interaction if the penetrance matrix could not be decomposed according to Risch's [19] definition of genetic heterogeneity. That is, formally, let P be 3×3 binary penetrance matrix. Then *P* is *not* an interaction if and only if there exists two 3×3 binary matrices, *R* with identical rows, and *C* with identical columns, and *P* cannot be written as the logical OR between R and C. The genetic heterogeneity definition was chosen because it excludes most marginal effect-only models, thereby reducing noise in the power estimation, and because it can easily be evaluated for complete penetrance matrices. The penetrance matrix was then reduced to continuous values by changing the 0's to a specified base risk of β_0 and the 1's to β_0 + β_1 . To enhance comparison of models, we used heritability, H^2 , as a summary measure of all genetic effects in a model, where heritability was defined as

$$H^{2} = \sum_{i,j} \frac{(\Pr(Y=1) - \Pr(Y=1 \mid X_{1}=i, X_{2}=j))^{2} \Pr(X_{1}=i, X_{2}=j)}{\Pr(Y=1)(1 - \Pr(Y=1))}$$

For each model, the parameter β_1 was adjusted to obtain heritabilities of 0.005, 0.010 and 0.015. Using this enumeration we obtain a set of models, each defined by a matrix of pene-trances for each genotype combination $Pr(Y = 1 | X_1 = i, X_2 = j)$ (cf. <u>S7</u>, <u>S8</u> and <u>S9</u> Figs and <u>S2</u> <u>Text</u>). The genotypes for cases and controls were then generated using Bayes' theorem

$$\Pr(X_1 = i, X_2 = j \mid Y = 1) = \frac{\Pr(Y = 1 \mid X_1 = i, X_2 = j) \Pr(X_1 = i) \Pr(X_2 = j)}{\sum_{k,l} \Pr(Y = 1 \mid X_1 = k, X_2 = l) \Pr(X_1 = k) \Pr(X_2 = l)}$$

to get the multinomial distribution over genotypes. We generated 1,000 data sets from each of these models. We assumed a balanced design (i.e., same number of cases and controls), the sample size for each group was varied over 2,000, 3,000, and 4,000, the heritability was varied over 0.015, 0.020 and 0.025, and the minor allele frequency was fixed to 0.3. Each data set

comprised a single interacting variant pair, and to model multiple testing, we assumed that there were 10⁶ variants and 10¹² variant pairs tested. For each model and each parameter combination, the power of a method to detect interaction was estimated over the 1,000 replicates. The method's power over the spectrum of tested interaction models were then summarized in an exceedence plot.

We performed two additional power analyses using a second simulation strategy, where we used data simulated from a specific interaction model, the double dominant model (described in <u>S2 Text.1</u>), in which $\alpha = \beta_1 = \beta_2 = \gamma_1 = \gamma_2 = 0$ and $\delta_{11} = \delta_{12} = \delta_{21} = \delta_{22} = x$, and a logit link function was used. The value *x* was then varied to get the heritability 0.01, 0.02 and 0.03. This analysis used a fixed sample size of 3,000 cases and 3,000 controls. The minor allele frequency was set to 0.3 at both loci.

In the first of these two power analyses, we investigated how the relative power in detecting an associated variant pair generated under an interaction model varies over the different individual stages in the stage-wise approach, specifically we compared the power in the first and the last stages (i.e., using the null models H_1 and H_4). The parameters of the double dominant model can be seen in <u>S6 Table</u>.

In the second power analysis, where we studied the power of the *static* method to detect an interacting pair as a function of the estimated number of marginally associated variants, we set the total number of variants tested, N = 1,000,000, and the number of marginally associated variants was varied, $M \in \{10, 20, 100\}$. For the *static* method the corrections for each stage, in order, then was set to N(N - 1)/2, $M \cdot N$, $M \cdot N$ and $M \cdot (M - 1)/2$. The parameters of the double dominant model can be seen in S7 Table.

Generation of synthetic data for FWER estimation

The FWER estimation was based on simulated data. We generated data from ten different null models representing two different cases: The first case corresponds to the null models in our stage-wise methodology: no association, *single main* association and five null models with *double main* effects corresponding to the link functions in Table 1; the second case represents a more challenging scenario and comprise three null models with multiple main effects.

For each null model we generated 200 data set replicates that contained 500 - L unassociated and *L* associated variants, where *L* depends on the null model, L = 0 for the no association, L = 1 for the single association, L = 2 for the double main models, and L = 10, 20, 30 for the multivariate model. For each data set there was therefore *L* associated variants according to the null model. For each replicate this resulted in 124,750 pairs. The minor allele frequency was sampled uniformly between 0.2 and 0.4. We sampled individuals until we obtained 4000 cases and 4000 controls. The parameters used in each null model can be seen in <u>S8 Table</u>.

For the null models with multiple main effects, we used an additive logistic regression model to generate the phenotype. Let $L \in \{10, 20, 30\}$ be the number of variants to include from the chromosome, then the model was defined

$$\log\left(\frac{p}{1-p}\right) = \beta_0 + \sum_{i=1}^{L} \beta_i x_i$$

where $x_i \in \{0, 1, 2\}$ and $\beta_i \sim N(0.15, 0.01)$. The intercept was set to -9.0.

Comparison of statistical methods

We evaluated the power of our *static* stage-wise scale-invariant method in comparison to six other statistical methods. In our main, large-scale analysis using data generated from an

enumeration of all possible interaction models (see Material and methods section Generation of synthetic data for estimation of statistical power), we restricted ourselves to statistical methods that could efficiently compute a p-value with enough precision to test how they performed in realistic scenarios: Four methods based on a direct interaction test (i.e., in our framework description above, testing hypothesis H_4 against H_A) with a logistic link function GLM, but employing different screening strategies: Logistic—no screening. Marginal+logistic—the marginal screening method described by [29], which uses a GLM that tests the marginal effect of each variant at an optimistic significance level 0.1 for screening. The screening approaches used in CSS+logistic [30] and \mathbb{R}^2 +logistic [31] are both LD_{cohort} -based, but differ in the definition of the χ^2 -based statistic, and the choice of significance threshold used for the screening: χ^2 \geq 3 (corresponding to $p \leq 0.39$) and $p \leq 10^{-4}$, respectively. Thirdly, we test two methods based on the LD-contrast test with different filtering strategies: LD-contrast—no screening. Six*pac*—the method of [23], which recodes variant genotypes into two binary variables (according to dominant and recessive coding) and then combines LD_{cases} screening with a LD-contrast main test. The significance level was set to 0.05. We assumed that there were 10^{12} variant pairs present on the chip and that there existed one interacting pair. For the methods without screening (Logistic and LD-contrast), as well as for the Sixpac method, we corrected for 10^{12} pairs. For the remaining screening methods, we corrected for the expected number of null variant pairs passing the screening, by taking the product of the p-value threshold and the total number of pairs (i.e., Marginal + logistic : $(0.1 \cdot 10^6) \approx 5 \cdot 10^9$, CSS+logistic: $0.39 \cdot 10^{12} = 3.9 \cdot 10^{11}$, and R^2 +logistic: $10^{-4} \cdot 10^{12} = 10^8$). A pair was declared significant if it passed the significance level of both the screening and the main test. For all these methods we used the Holm-Bonferroni correction for multiple testing, which is more powerful than the classic Bonferroni correction. For the *Static* stage-wise method we corrected for 10^{12} pairs, $100 \cdot 10^{6}$ pairs, $100 \cdot 10^{6}$ pairs and 4950 pairs in each of the four stages respectively, to simulate the situation with 100 associated variants.

We also performed a second, smaller-scaled, but computationally more demanding, power comparison using data generated from specific interaction models and null models (described in detail in <u>S2 Text</u>). In addition to the seven methods enumerated above, this comparison also included our *adaptive* stage-wise, scale-invariant method and the Model-Based MDR (*MB-MDR*) method [26], which is a parametric extension of the MDR method that addresses some shortcomings of the original MDR method, in particular adjustment for main effects (these methods require the generation of data sets complete with both null and interaction pairs and could not be evaluated in the main power comparison above).

Lastly, we also used the same nine methods in a FWER comparison using simulated data generated as described in Material and methods section Generation of synthetic data for FWER estimation.

Software availability

A C++ implementation of all methods and source code for all experiments is available at: <u>https://github.com/mfranberg/besiq</u>.

Supporting Information

S1 Text. Proofs. Mathematical proofs for the FWER control of the *static* and *adaptive* method and the proof that the likelihood of models H_1 , H_2 , H_3 and H_A is invariant of the link function. (PDF)

S2 Text. Supplemental simulations, analyses, and methods. Supplemental power-analyses using simulated data generated from specific interaction models and from null models (shown in <u>S1–S4</u> Figs). Parameter information for all simulations for power analyses (cf., e.g., <u>S7–S8</u> Figs). Methods used in the investigation of the impact of weights in the stage-wise methodology (<u>S2 Table</u>). Implementation details of published methods for detection of interaction, used in all power and FWER comparisons. (PDF)

S1 Fig. Statistical power as a function of sample size and minor allele frequency under the double dominant model. The x-axis is the effect level (i.e., categorized effect sizes, cf. <u>S2 Text</u>), and the y-axis is the power. The columns correspond to different minor allele frequencies. The rows correspond to different sample sizes under a balanced design e.g. 2000 indicates 2000 cases and 2000 controls. (Notice that the red line for the *adaptive* method is often hidden behind the black line for the *static* method.) (TIF)

S2 Fig. Statistical power as a function of sample size and minor allele frequency under the double recessive model. The x-axis is the effect level (i.e., categorized effect sizes, cf. <u>S2 Text</u>), and the y-axis is the power. The columns correspond to different minor allele frequencies. The rows correspond to different sample sizes under a balanced design e.g. 2000 indicates 2000 cases and 2000 controls. (Notice that the red line for the *adaptive* method is often hidden behind the black line for the *static* method.) (TIF)

S3 Fig. Statistical power as a function of sample size and minor allele frequency under the XOR model. The x-axis is the effect level (i.e., categorized effect sizes, cf. <u>S2 Text</u>), and the y-axis is the power. The columns correspond to different minor allele frequencies. The rows correspond to different sample sizes under a balanced design e.g. 2000 indicates 2000 cases and 2000 controls. (Notice that the red line for the *adaptive* method is often hidden behind the black line for the *static* method.) (TIF)

S4 Fig. Statistical power as a function of sample size and minor allele frequency under the side model. The x-axis is the effect level (i.e., categorized effect sizes, cf. <u>S2 Text</u>), and the y-axis is the power. The columns correspond to different minor allele frequencies. The rows correspond to different sample sizes under a balanced design e.g. 2000 indicates 2000 cases and 2000 controls.

(TIF)

S5 Fig. Estimated penetrances for the rs602633 (PSRC1) and rs4694178 (IL8, CXCL6) interaction in the discovery cohort. The x-axis is the number of minor alleles of the first variant and the y-axis is the estimated penetrance under the different models. The colors correspond to the number of minor alleles of the second variant. The error bars indicate the estimated confidence interval. (EPS)

S6 Fig. Estimated penetrances for the rs602633 (PSRC1) and rs4694178 (IL8, CXCL6) interaction in the replication cohort. The x-axis is the number of minor alleles of the first variant and the y-axis is the estimated penetrance under the different models. The colors correspond to the number of minor alleles of the second variant. The error bars indicate the estimated confidence interval. (EPS) S7 Fig. The estimated density of effect sizes for models evaluated in the main power experiment represented on the logit scale. The rows correspond to sample sizes, and the columns to heritabilities. The differently colored lines correspond to the estimated density of different parameters in the models. The label "a" refers to the intercept α , "b1" and "b2" to the main effects of the first variant β_1 and β_2 , "c1" and "c2" to the main effects of the second variant γ_1 and γ_2 , and "d11", "d12", "d21" and "d22" to the interaction effects δ_{11} , δ_{12} , δ_{21} and δ_{22} . (TIF)

S8 Fig. The estimated density of effect sizes for models evaluated in the main power experiment represented on the penetrance scale. The rows correspond to sample sizes, and the columns to heritabilities. The differently colored lines correspond to the estimated density of different parameters in the models. The label "a" refers to the intercept α , "b1" and "b2" to the main effects of the first variant β_1 and β_2 , "c1" and "c2" to the main effects of the second variant γ_1 and γ_2 , and "d11", "d12", "d21" and "d22" to the interaction effects δ_{11} , δ_{12} , δ_{21} and δ_{22} . (TIF)

S9 Fig. The cumulative distribution function of marginal heritability fraction for the main power experiment. The x-axis represents a threshold for the marginal heritability fraction, and y-axis the fraction models with a marginal heritability fraction greater than this threshold. The blue line represents the empirical cumulative distribution function. The black line represents the cumulative distribution for a uniform distribution. The marginal heritability fraction was computed as the sum of the marginal heritabilities divided by the total heritability. (TIF)

S1 Table. Robustly associated variants genotyped on the IBC chip. A list of the top variants in each of the published CAD loci associated in large-scale meta analyses [2], that are also present on the IBC-chip.

(PDF)

S2 Table. Parameters, for the CAD-associated variant pairs, estimated under a logistic regression model. Variant pairs that displayed significant interaction on at least one scale for each strategy in the original analysis of the PROCARDIS cohort are listed. The intercept is α . The main effects are β_1 , β_2 , γ_1 , and γ_2 . The interaction parameters are δ_{11} , δ_{12} , δ_{21} , and δ_{22} . LR is the likelihood ratio which measures the degree of evidence for the interaction model compared to the additive model on the logistic scale. (PDF)

S3 Table. Parameters, for the CAD-associated variant pairs, estimated under a logistic regression model after adjustment for population stratification (3 MDS components), age, sex and smoking. Variant pairs that displayed significant interaction on at least one scale for each strategy in the original analysis of the PROCARDIS cohort are listed. The intercept is α . The main effects are β_1 , β_2 , γ_1 , and γ_2 . The interaction parameters are δ_{11} , δ_{12} , δ_{21} , and δ_{22} . LR is the likelihood ratio which measures the degree of evidence for the interaction model compared to the additive model on the logistic scale. (PDF)

S4 Table. The genotype frequencies of the significant variant pairs for the discovery and replication analyses. The MAF1 and MAF2 columns denote the minor allele frequency for variant 1 and variant 2 respectively. The counts n_{ij} is the number of individuals with genotype *i* at the first variant and genotype *j* at the second variant. (PDF)

S5 Table. Evaluation of how different weight combinations affect the statistical power of the *static* and *adaptive* stage-wise methods. The weight for a stage $i \in [4]$ is indicated by w_i . The list is sorted in ascending order by the power obtained for the *static* stage-wise method and the weight combination selected for all remaining analyses in this paper is indicated by bold face.

(PDF)

S6 Table. The penetrance matrices used for the double dominant model when comparing the statistical power between stages. The value p_{ij} denotes the penetrance for genotype ij. The rows correspond to the sequence of effect sizes used, sorted in ascending order of the effect size.

(PDF)

S7 Table. The penetrance matrices used for the double dominant model when investigating the statistical power for the static method. The value p_{ij} denotes the penetrance for genotype ij. The rows correspond to the sequence of effect sizes used, sorted in ascending order of the effect size.

(PDF)

S8 Table. Parameters used for the generation of simulated data from seven different GLMs used to estimate FWER. The GLMS corresponds to the null models used in the stage-wise methodology. The intercept is α . The main effects are β_1 , β_2 , γ_1 , and γ_2 . (PDF)

Acknowledgments

John Öhrvik provided valuable discussions and comments to an earlier version of the manuscript. Martin Farrall, Anuj Goel and Hugh Watkins made suggestions relating to the biological data. We also would like to thank the editor and two anonymous reviewers whose suggestions greatly improved the quality of this manuscript.

We also thank all the members of the PROCARDIS consortium:

Project Steering Committee: Hugh Watkins (chair)^{1,2}, Anders Hamsten³, Rory Collins⁴, Udo Seedorf⁵, Maria Grazia Franzosi⁶, Jörg Hager⁷, Fiona Green⁸, Michael Parker⁹, Jose Manuel Soria¹⁰, Elena Tremoli^{11,12}, Johan Björkegren¹³, Anna Walentinsson¹⁴, Carla Finocchiaro¹⁵

Department of Cardiovascular Medicine, University of Oxford: Anuj Goel¹, Martin Farral^{1,2}, Shapour Jalilzadeh¹, Theodosios Kyriakou¹, Halit Ongen¹, John F Peden¹, Hugh Watkins^{1,2}

Atherosclerosis Research Unit, Department of Medicine, Solna, Karolinska Institutet: Per Eriksson³, Lasse Folkersen³, Anders Hamsten³, Mai-Lis Hellénius³, Ferdinand vant Hooft³, Jacob Lagercrantz³, Anders Mälarstig³, Maria Sabater-Lleal³, Bengt Sennblad³, Angela Silveira³, Rona Strawbridge³, John Öhrvik³

Clinical Trial Service Unit, University of Oxford: Robert Clarke⁴, Rory Collins⁴, Jemma C Hopewell⁴, Pamela Linksted⁴, Sarah Parish⁴

Leibniz-Institut für Arterioskleroseforschung an der Universität Münster: Gerd Assmann⁵, Stephan Rust⁵, Udo Seedorf⁵

Departmentof Cardiovascular Research, Istituto Mario Negri: Simona Barlera⁶, Maria Grazia Franzosi⁶, Gianni Tognoni⁶

Centre National de Genotypage: Marc Delepine⁷, Jörg Hager⁷, Simon Heath⁷, Mark Lathrop⁷

Division of Biomedical Sciences, University of Surrey: Fiona Green⁸

The Ethox Centre, University of Oxford: Paula Boddington⁹, Michael Parker⁹

Institut de Recerca del Hospital de la Santa Creu I Sant Pau: Alfonso Buil¹⁰, Jose Manuel Soria¹⁰, Juan Carlos Souto¹⁰

Centro Cardiologico Monzino and Dipartimento di Scienze Farmacologiche e Biomolecolari, University of Milan: Cristina Banfi^{11,12}, Elena Tremoli^{11,12}

Clinical Gene Networks AB: Johan Björkegren¹³, Josefin Skogsberg¹³, Jesper Tegnér¹³

CV Genetics and Molecular Biology, AstraZeneca R&D: Per G Olsson¹⁴, Gunnar O Olsson¹⁴, Anna Walentinsson¹⁴

CF consulting S.r.l: Serena Cogoni¹⁵, Carla Finocchiaro¹⁵, Tessa Gronchi¹⁵

- Department of Cardiovascular Medicine, The Wellcome Trust Centre for Human Genetics, University of Oxford, Oxford, United Kingdom;
- Department of Cardiovascular Medicine, University of Oxford, John Radcliffe Hospital, Headington, Oxford, United Kingdom;
- 3. Atherosclerosis Research Unit, Department of Medicine, Solna, Karolinska Institutet, Karolinska University Hospital Solna, Stockholm, Sweden;
- 4. Clinical Trial Service Unit, University of Oxford, United Kingdom;
- 5. Leibniz-Institut für Arterioskleroseforschung an der Universität Münster, Münster, Germany;
- 6. Centre National de Genotypage, Institut Genomique, Commisariat L'Energie Atomique, Evry, France;
- 7. Department of Cardiovascular Research, Istituto Mario Negri, Milan, Italy;
- 8. Division of Biomedical Sciences, Faculty of Health and Medical Sciences, University of Surrey, Surrey, United Kingdom;
- 9. The Ethox Centre, Institute of Health Sciences, University of Oxford, Oxford, United Kingdom;
- 10. Institut de Recerca del Hospital de la Santa Creu I Sant Pau, Barcelona, Spain;
- 11. Centro Cardiologico Monzino, IRCCS, Milan, Italy;
- 12. Dipartimento di Scienze Farmacologiche e Biomolecolari, University of Milan, Milan, Italy;
- 13. Clinical Gene Networks AB, Stockholm, Sweden;
- 14. CV Genetics and Molecular Biology, AstraZeneca R&D, Mölndal, Sweden;
- 15. CF consulting S.r.l, Milan, Italy.

Author Contributions

Conceived and designed the experiments: MF BS JL. Performed the experiments: MF. Analyzed the data: MF BS KG JL. Contributed reagents/materials/analysis tools: AH. Wrote the paper: KG MF BS JL.

References

1. GBD 2013 Mortality and Causes of Death Collaborators. Global, regional, and national age-sex specific all-cause and cause-specific mortality for 240 causes of death, 1990–2013: a systematic analysis for the Global Burden of Disease Study 2013. The Lancet. 2015 January; 385(9963):117–171.

- CARDIoGRAMplusC4D Consortium, Deloukas P, Kanoni S, Willenborg C, Farrall M, Assimes TL, et al. Large-scale association analysis identifies new risk loci for coronary artery disease. Nat Genet. 2013 January; 45(1):25–33. doi: <u>10.1038/ng.2480</u> PMID: <u>23202125</u>
- Manolio TA, Collins FS, Cox NJ, Goldstein DB, Hindorff LA, Hunter DJ, et al. Finding the missing heritability of complex diseases. Nature. 2009 October; 461(7265):747–53. doi: <u>10.1038/nature08494</u> PMID: <u>19812666</u>
- Lehner B. Genotype to phenotype: lessons from model organisms for human genetics. Nat Rev Genet. 2013 January; 14(3):168–178. doi: <u>10.1038/nrg3404</u> PMID: <u>23358379</u>
- Bateson W, Saunders E, Punnett R. Experimental studies in the physiology of heredity. Roy Soc Evolution Com Rpts. 1905; 2:1–55.
- Kelley R, Ideker T. Systematic interpretation of genetic interactions using protein networks. Nat Biotechnol. 2005 May; 23(5):561–6. doi: <u>10.1038/nbt1096</u> PMID: <u>15877074</u>
- McLellan J, O'Neil N, Tarailo S, Stoepel J, Bryan J, Rose A, et al. Synthetic Lethal Genetic Interactions That Decrease Somatic Cell Proliferation in Caenorhabditis elegans Identify the Alternative RFCCTF18 as a Candidate Cancer Drug Target. Mol Biol Cell. 2009 October; 20:5305–5313. doi: <u>10.1091/mbc.</u> <u>E09-08-0699</u>
- Szappanos B, Kovács K, Szamecz B, Honti F, Costanzo M, Baryshnikova A, et al. An integrated approach to characterize genetic interaction networks in yeast metabolism. Nat Genet. 2011 May; 43:656–662. doi: 10.1038/ng.846 PMID: 21623372
- Leamy LJ, Pomp D, Lightfoot JT. An Epistatic Genetic Basis for Physical Activity Traits in Mice. J Hered. 2008 May; 99(6):639–646. doi: <u>10.1093/jhered/esn045</u> PMID: <u>18534999</u>
- Gaertner BE, Parmenter MD, Rockman MV, Kruglyak L, Phillips PC. More than the sum of its parts: a complex epistatic network underlies natural variation in thermal preference behavior in Caenorhabditis elegans. Genetics. 2012 December; 192(4):1533–1542. doi: <u>10.1534/genetics.112.142877</u> PMID: 23086219
- Huang W, Richards S, Carbone MA, Zhu D, Anholt RRH, Ayroles JF, et al. Epistasis dominates the genetic architecture of Drosophila quantitative traits. Proc Natl Acad Sci U S A. 2012 September; 109 (39):15553–15559. doi: 10.1073/pnas.1213423109 PMID: 22949659
- Gibson G. Epistasis and pleiotropy as natural properties of transcriptional regulation. Theor Popul Biol. 1996 January; 49:58–89. doi: 10.1006/tpbi.1996.0003 PMID: 8813014
- Cordell HJ. Detecting gene-gene interactions that underlie human diseases. Nat Rev Genet. 2009 June; 10(6):392–404. doi: 10.1038/nrg2579 PMID: 19434077
- Steen KV. Travelling the world of gene-gene interactions. Brief Bioinform. 2012 January; 13(1):1–19. doi: <u>10.1093/bib/bbr012</u> PMID: <u>21441561</u>
- Sham PC, Purcell SM. Statistical power and significance testing in large-scale genetic studies. Nat Rev Genet. 2014 April; 15(5):335–346. doi: 10.1038/nrg3706 PMID: 24739678
- Cordell HJ. Epistasis: what it means, what it doesn't mean, and statistical methods to detect it in humans. Hum Mol Genet. 2002 October; 11(20):2463–8. doi: <u>10.1093/hmg/11.20.2463</u> PMID: 12351582
- Clayton D. Link functions in multi-locus genetic models: implications for testing, prediction, and interpretation. Genet Epidemiol. 2012 May; 36(4):409–18. doi: 10.1002/gepi.21635 PMID: 22508388
- Good IJ. Maximum Entropy for Hypothesis Formulation, Especially for Multidimensional Contingency Tables. Ann Math Stat. 1963 March; 34:911–934. doi: 10.1214/aoms/1177704014
- Risch N. Linkage Strategies for Genetically Complex Traits. 1. Multilocus Models. Am J Hum Genet. 1990 February; 46(2):222–228. PMID: 2301392
- 20. Loftus GR. On interpretation of interactions. Mem Cognit. 1978 February; 6:312–319. doi: 10.3758/ BF03197461
- Knol MJ, VanderWeele TJ. Recommendations for presenting analyses of effect modification and interaction. Intl J Epidemiol. 2012 April; 41(2):514–20. doi: <u>10.1093/ije/dyr218</u>
- Nielsen DM, Ehm MG, Zaykin DV, Weir BS. Effect of two- and three-locus linkage disequilibrium on the power to detect marker/phenotype associations. Genetics. 2004 October; 168(2):1029–1040. doi: <u>10.</u> <u>1534/genetics.103.022335</u> PMID: <u>15514073</u>
- Prabhu S, Pe'er I. Ultrafast genome-wide scan for SNP-SNP interactions in common complex disease. Genome Res. 2012 November; 22(11):2230–2240. doi: <u>10.1101/gr.137885.112</u> PMID: <u>22767386</u>
- Ritchie MD, Hahn LW, Roodi N, Bailey LR, Dupont WD, Parl FF, et al. Multifactor-dimensionality reduction reveals high-order interactions among estrogen-metabolism genes in sporadic breast cancer. Am J Hum Genet. 2001; 69(1):138–147. doi: 10.1086/321276 PMID: 11404819

- Chung Y, Lee SY, Elston RC, Park T. Odds ratio based multifactor-dimensionality reduction method for detecting gene-gene interactions. Bioinformatics. 2007 January; 23(1):71–76. doi: <u>10.1093/</u> <u>bioinformatics/btl557</u> PMID: <u>17092990</u>
- Calle ML, Urrea V, Vellalta G, Malats N, Van Steen K. Model-Based Multifactor Dimensionality Reduction for detecting interactions in high-dimensional genomic data. Technical Reports, Department of Systems Biology, Universitat de Vic. 2008 January;p. 1–14. Available from: <u>http://repositori.uvic.cat/</u> bitstream/handle/10854/408/docrec_a2008_calle_malu_mbmdr.pdf?sequence=1.
- Wan X, Yang C, Yang Q, Xue H, Tang NLS, Yu W. Detecting two-locus associations allowing for interactions in genome-wide association studies. Bioinformatics. 2010 October; 26(20):2517–25. doi: <u>10.</u> <u>1093/bioinformatics/btq486</u> PMID: <u>20736343</u>
- Ueki M, Cordell HJ. Improved statistics for genome-wide interaction analysis. PLoS Genet. 2012 January; 8(4):e1002625. doi: <u>10.1371/journal.pgen.1002625</u> PMID: <u>22496670</u>
- Marchini J, Donnelly P, Cardon LR. Genome-wide strategies for detecting multiple loci that influence complex diseases. Nat Genet. 2005 April; 37(4):413–7. doi: <u>10.1038/ng1537</u> PMID: <u>15793588</u>
- Millstein J, Conti DV, Gilliland FD, Gauderman WJ. A testing framework for identifying susceptibility genes in the presence of epistasis. Am J Hum Genet. 2006 Jan; 78(1):15–27. doi: <u>10.1086/498850</u> PMID: <u>16385446</u>
- Lewinger JP, Morrison JL, Thomas DC, Murcray CE, Conti DV, Li D, et al. Efficient Two-Step Testing of Gene-Gene Interactions in Genome-Wide Association Studies. Genet Epidemiol. 2013 April; 37 (5):440–451. doi: 10.1002/gepi.21720 PMID: 23633124
- Marcus R, Eric P, Gabriel KR. On Closed Testing Procedures with Special Reference to Ordered Analysis of Variance. Biometrika. 1976 December; 63:655–660. doi: <u>10.1093/biomet/63.3.655</u>
- Wright SPW. Adjusted P-Values for Simultaneous Inference. Biometrics. 1992 December; 48:1005– 1013. doi: <u>10.2307/2532694</u>
- Berger R. Likelihood Ratio Tests and Intersection-Union Tests. In: Panchapakesan S, Balakrishnan N, editors. Advances in Statistical Decision Theory and Applications. Statistics for Industry and Technology. Birkhäuser Boston; 1997. p. 225–237. Available from: <u>http://dx.doi.org/10.1007/978-1-4612-</u> 2308-5_15.
- Li W, Reich J. A complete enumeration and classification of two-locus disease models. Hum Hered. 2000 December; 50(6):334–49. doi: <u>10.1159/000022939</u> PMID: <u>10899752</u>
- Lee I, Blom UM, Wang PI, Shim JE, Marcotte EM. Prioritizing candidate disease genes by networkbased boosting of genome-wide association data. Genome Res. 2011 July; 21:1109–1121. doi: <u>10.</u> <u>1101/gr.118992.110</u> PMID: <u>21536720</u>
- Hindorff LA, Sethupathy P, Junkins HA, Ramos EM, Mehta JP, Collins FS, et al. Potential etiologic and functional implications of genome-wide association loci for human diseases and traits. Proc Natl Acad Sci U S A. 2009 June; 106(23):9362–9367. doi: 10.1073/pnas.0903103106 PMID: 19474294
- Benjamini Y, Hochberg Y. Controlling the false discovery rate: a practical and powerful approach to multiple testing. J R Stat Soc Series B Stat Methodol. 1995 March; 57(1):289–300.
- Hamsten A, Eriksson P. Identifying the susceptibility genes for coronary artery disease: from hyperbole through doubt to cautious optimism. J Intern Med. 2008 May; 263:538–552. doi: <u>10.1111/j.1365-2796.</u> 2008.01958.x PMID: <u>18410597</u>
- 40. Arvind P, Nair J, Jambunathan S, Kakkar VV, Shanker J. CELSR2-PSRC1-SORT1 gene expression and association with coronary artery disease and plasma lipid levels in an Asian Indian cohort. J Cardiol. 2014 November; 64(5):339–346. doi: 10.1016/j.jjcc.2014.02.012 PMID: 24674750
- Libby P. Inflammation in atherosclerosis. Arterioscler Thromb Vasc Biol. 2012 September; 32(9):2045– 51. doi: 10.1161/ATVBAHA.108.179705 PMID: 22895665
- IBC 50K CAD Consortium, Butterworth AS, Braund PS, Farrall M, Hardwick RJ, Saleheen D, et al. Large-scale gene-centric analysis identifies novel variants for coronary artery disease. PLoS Genet. 2011 September; 7(9):e1002260. doi: <u>10.1371/journal.pgen.1002260</u>
- 43. Keating BJ, Tischfield S, Murray SS, Bhangale T, Price TS, Glessner JT, et al. Concept, design and implementation of a cardiovascular gene-centric 50 k SNP array for large-scale genomic association studies. PloS One. 2008 October; 3:e3583. doi: <u>10.1371/journal.pone.0003583</u> PMID: <u>18974833</u>
- Wang K, Li M, Hakonarson H. ANNOVAR: functional annotation of genetic variants from high-throughput sequencing data. Nucleic Acids Res. 2010 September; 38(16):e164. doi: <u>10.1093/nar/gkq603</u> PMID: <u>20601685</u>
- 45. Kinsella RJ, Kahari A, Haider S, Zamora J, Proctor G, Spudich G, et al. Ensembl BioMarts: a hub for data retrieval across taxonomic space. Database (Oxford). 2011 January; 2011(0):bar030–bar030. doi: 10.1093/database/bar030